Author's response to reviews

Title: Influence of bi- and tri-compartmental knee arthroplasty on the kinematics of the knee joint

Authors:

Markus Wünschel (Markus.Wuenschel@med.uni-tuebingen.de)
JiaHsuan Lo (Joshualo@umich.edu)
Torsten Dilger (Torsten.Dilger@kabelbw.de)
Nikolaus Wülker (Wuelker@med.uni-tuebingen.de)
Otto Müller (Otto.Mueller@med.uni-tuebingen.de)

Version: 5 Date: 27 January 2011

Author's response to reviews:

Some minor changes were made mostly concerning spelling.

Dr. M. Wünschel